Author's response to reviews

Title: Long-term cancer mortality time trends in Ohio, USA, 1970-2001: differences by race, gender and age

Authors:

Jerzy E Tyczynski (jerzytycz@yahoo.com)
Hans (J) Berkel (berkel@cancerpreventioninstitute.org)

Version: 5 Date: 5 October 2005

Author's response to reviews:

Requested changes have been implemented to graphs and tables.